Author’s response to reviews

Title: Decreased risk of falls in patients attending music sessions on an acute geriatric ward: results from a retrospective cohort study

Authors:

Julia Chabot (julia.chabot@mcgill.ca)

Olivier Beauchet (olivier.beauchet@mcgill.ca)

Shek Fung (shek.fung@mcgill.ca)

Isabelle Peretz (isabelle.peretz@umontreal.ca)

Version: 1  Date: 07 Nov 2017

Author’s response to reviews:

Thank you for your comments. Here are the comments as well as the changes which were done:

1. We would appreciate it if you could please provide us with the full, official name of the committee which granted you approval for this study, and committee reference number if appropriate. Additionally, please clarify that you had the written consent of all participates to participate in your study in this section.

The name of the ethics committee was added as well as the reference number of the study. As the study was done retrospectively and as participation to the invention was voluntary, no written consent was obtained.

2. Please refer to all authors by their initials in your Authors’ contribution section, and confirm that all authors have read and approved of the final manuscript in this section.

This modification was done.